Author's response to reviews

Title: Chinese women's knowledge and future preference after undergoing non-invasive prenatal testing for positive aneuploidy screening

Authors:

KO KOU (kouko@ha.org.hk)
CF POON (pcf301@ha.org.hk)
WC TSE (twc320@pmh.ha.org.hk)
SL MAK (msl061@ha.org.hk)
KY LEUNG (leungky1@ha.org.hk)

Version: 2 Date: 7 February 2015

Author's response to reviews: see over
7 February 2015

Mr. Jason Pepito
Journal Editorial Office
BioMed Central

Dear Mr. Jason Pepito

Chinese women’s knowledge and future preference after undergoing non-invasive prenatal testing for positive aneuploidy screening KO KOU, CF POON, WC TSE, SL MAK and KY LEUNG BMC Pregnancy and Childbirth (Section: Trends in pregnancy and childbirth)

Many thanks for processing our manuscript and valuable advice. We have revised our manuscript accordingly.

1. We have added line numbering.
2. We have added email addresses of all coauthors.
3. We have added acknowledgment.
4. We have included all the tables after the references instead of as additional files.

We have also changed the format of our manuscript according to the requirements of your journal, corrected the typo, grammatical errors, and revised some wordings of our manuscript.

We hope that our manuscript now fulfills the requirements from your journal, and are prepared to further revise our manuscript if necessary.

Best Regards,

KY
Dr. KY Leung
Chief of Service, Department of O&G, Queen Elizabeth Hospital
Chairman, Ultrasound Committee, Asia Oceania Federation of Obstetrics and Gynaecology
Ambassador, International Society of Ultrasound in Obstetrics and Gynaecology
Member, International Fetoscopy Group